Author’s response to reviews

Title: Therapeutic and immunomodulatory activities of short-course treatment of murine visceral leishmaniasis with KALSOMETM10, a new liposomal amphotericin B

Authors:

Md Asad (asadseraphic.nii@gmail.com)
Pradyot Bhattacharya (b_pradyot@yahoo.com)
Nahid Ali (nali@iicb.res.in)
Antara Banerjee (banerjee.antara77@gmail.com)

Version: 4 Date: 16 January 2015

Author’s response to reviews:

Dear Sir,

Please find attached our revised manuscript entitled “Therapeutic and immunomodulatory activities of a short-course treatment of murine visceral leishmaniasis with KALSOMETM10, a new liposomal amphotericin B” by Md Asad, Pradyot Bhattacharya, Antara Banerjee and Nahid Ali for consideration of publication in your esteemed journal. Here we have incorporated all the necessary corrections and improvement of the manuscript as suggested by the reviewers (three new figures and one table in addition to the earlier submission). We are highly thankful to the editorial board as well as the reviewers for critically reviewing our manuscript and helping us in improving it scientifically.

I hope this paper will generate interest in the area of drug delivery and its further evaluation will lead to development of a safe, low dose and cheap therapy for visceral leishmaniasis.

Thanking you,

Yours sincerely,

Nahid Ali